

Northward Expansion and Intensification of Phytoplankton Growth During the Early Ice-Free Season in Arctic

S. Renaut, E. Devred, M. Babin

► To cite this version:

S. Renaut, E. Devred, M. Babin. Northward Expansion and Intensification of Phytoplankton Growth During the Early Ice-Free Season in Arctic. Geophysical Research Letters, 2018, 45 (19), pp.10,590-10,598. 10.1029/2018GL078995 . hal-02415146

HAL Id: hal-02415146 https://hal.science/hal-02415146

Submitted on 10 Apr2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License



Geophysical Research Letters

RESEARCH LETTER

10.1029/2018GL078995

Key Points:

- We report on the temporal and spatial variability of phytoplankton spring blooms between 2003 and 2013 in the Arctic Ocean
- Primary productivity of phytoplankton spring blooms is increasing in several areas of the Arctic Ocean between 2003 and 2013
- Satellite observations revealed a northward expansion of these blooms at a rate of 1° per decade

Supporting Information:

Supporting Information S1

Correspondence to: S. Renaut, sophie.renaut@takuvik.ulaval.ca

Citation:

Renaut, S., Devred, E., & Babin, M. (2018). Northward expansion and intensification of phytoplankton growth during the early ice-free season in Arctic. *Geophysical Research Letters*, 45, 10,590–10,598. https://doi.org/10.1029/ 2018GL078995

Received 31 MAY 2018 Accepted 23 AUG 2018 Published online 15 OCT 2018

©2018. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Northward Expansion and Intensification of Phytoplankton Growth During the Early Ice-Free Season in Arctic

S. Renaut¹ (D), E. Devred^{1,2}, and M. Babin¹ (D)

¹Takuvik Joint International Laboratory, ULaval (Canada) & CNRS (France), Département de Biologie, and Québec-Océan, Université Laval, Québec, Canada, ²Ecosystem and Ocean Sciences Division, Bedford Institute of Oceanography, Department of Fisheries and Oceans, Dartmouth, Nova Scotia, Canada

Abstract In the last decades, reduction of the ice cover has been documented to affect the structure and the functioning of Arctic marine ecosystems. One direct consequence is earlier phytoplankton spring blooms and larger annual primary production compared with previous decades. However, the impact of changes in the dynamics of sea ice specifically on phytoplankton spring blooms, a major contributor of the annual primary production in the Arctic Ocean, remains poorly known. Here we report on their temporal and spatial variabilities in open waters between 2003 and 2013 using satellite ocean color data. We observed a significant increase in primary productivity of phytoplankton spring blooms in different sectors of the Arctic Ocean, especially in the Barents and Kara Seas. Satellite observations also revealed a northward expansion of these blooms at a rate of 1° per decade driven by the Barents and the Kara regions.

Plain Language Summary The declining of the ice cover in the Arctic Ocean has a strong impact on the dynamics of the marine ecosystem. The phytoplankton spring bloom, which can contribute to more than half of the annual primary production in open waters in some regions of the Arctic Ocean and which is tightly linked to the ice cover, is undergoing drastic changes. This key feature represents also a significant food source for higher trophic levels. Our study reveals rapid changes in the magnitude of primary productivity and the spatial distribution of these blooms. We observed a significant increase in primary productivity of phytoplankton spring blooms in different areas of the Arctic Ocean. A northward expansion of these blooms was also observed. Their occurrence at very high latitudes in the central basin, where they were nonexistent and where a low productivity usually prevails, is also a significant finding.

1. Introduction

The shrinking of the summer ice pack in the Arctic Ocean (Stroeve et al., 2012) is reshaping marine ecosystems (Post et al., 2013; Serreze et al., 2007). Annual and summer monthly maximum pan-Arctic net primary production have increased by approximately 47% between 1998 and 2015 in the spring and summer months, as a result of the increase in surface area of open waters during summer and a longer phytoplankton growth season (Kahru et al., 2016). In spring, the growth of primary producers in open waters greatly depends on sea ice dynamics (Sakshaug & Skjoldal, 1989). Following the ice breakup, favorable light and nutrient conditions generally support phytoplankton blooms along the ice edge at moderate polar latitudes (Tremblay & Gagon, 2009). A recent study showed that the length of the productive period has increased by about 17 days between 1998 and 2015 in open waters of the Arctic Ocean due to an earlier ice retreat and a later freezing (Kahru et al., 2016). Additionally, the phytoplankton bloom may start, and even end, more frequently under sea ice before breakup (Arrigo et al., 2012; Lowry et al., 2014; Mundy et al., 2009, 2014), since the ice pack may increasingly allow light to penetrate the ocean during spring due to a decrease in snow precipitation, thinner ice (Haas et al., 2008), and more abundant leads and melt ponds (Assmy et al., 2017). The frequency of fall blooms, a common trait at temperate latitudes, is also increasing at high latitudes (>70°N) (Ardyna et al., 2014). Overall, the phytoplankton phenology of this polar ecosystem has undergone drastic changes in recent decades and is likely to keep evolving in the future.

Despite the short duration of the phytoplankton spring bloom (PSB; i.e., a few weeks), it can contribute more than half of the annual primary production in some regions of the Arctic Ocean (Perrette et al., 2011; Sakshaug, 2004) and represents a significant food source for higher trophic levels (Wassmann et al., 2008). Until recently, at very high latitudes (i.e., central basin, >80°N), PSBs were nonexistent, and productivity was low due to the presence of the perennial sea ice and the late breakup of first-year sea ice in September while low solar



radiation prevails (Kröncke et al., 2000). Several studies using ocean color remote sensing have investigated recent multiyear changes in annual primary production with satellite observations in the Arctic Ocean (Arrigo et al., 2008; Arrigo & van Dijken, 2011, 2015; Babin et al., 2015; Bélanger et al., 2013; Matrai et al., 2013; Pabi et al., 2008; Petrenko et al., 2013). Also, Kahru et al. (2011) showed that chlorophyll *a* (chl *a*) concentration is increasingly peaking earlier in the year in several regions of the Arctic Ocean. More recently, Kahru et al. (2016) documented a 47% increase in the summer monthly maximum pan-Arctic primary production. But the impact of ongoing changes in Arctic sea ice specifically on the PSB has not yet been addressed with a dedicated regional approach. Understanding the dynamics of PSBs, a key feature of the open water season in the Arctic, is critical given its preponderant role in sustaining the entire marine ecosystem. In this study, we used satellite observations of ocean color, which provide estimates of phytoplankton biomass and primary productivity (PP), to assess the dynamics of PSBs between 2003 and 2013 in the Arctic Ocean. Our approach was inspired by the study of Perrette et al. (2011) who showed, using ocean color data from year 2007, that the PSB contributes a significant fraction of annual primary production in the Arctic Ocean.

2. Methods

2.1. Satellite Data

Daily binned Level-3 (4.64 km spatial resolution) chl *a* concentrations, an index of phytoplankton biomass, from the Moderate-Resolution Imaging Spectroradiometer Sensor (MODIS, 2003–2013) were downloaded from NASA's ocean color website (http://oceandata.sci.gsfc.nasa.gov). The GSM01 semianalytical bio-optical algorithm was used to retrieve chl *a* concentration from remote sensing reflectance (R_{rs} ; Maritorena & Siegel, 2005). This algorithm was selected because it was found to perform better than standard empirical algorithms, such as the standard 3-band OC3 MODIS ocean color algorithm, in Arctic waters dominated by colored dissolved organic matter (Ben Mustapha et al., 2012). Also, satellite observations of surface chl *a* may not correlate well with the depth-integrated biomass when a subsurface chlorophyll maximum is present, which can happen in some regions of the Arctic Ocean, close to the nitracline, when surface nutrients become depleted (Martin et al., 2010). However, previous sensitivity analyses that used large data sets of in situ vertical chl *a* profiles suggest a limited impact of subsurface chlorophyll maximums on large-scale estimates of depth-integrated PP in the Arctic Ocean (Ardyna et al., 2013; Arrigo et al., 2011).

Daily sea ice cover and concentrations, downloaded from the National Snow and Ice Data Center (http:// nsidc.org) and derived from the Special Sensor Microwave/Imager (2003–2007) and from the Special Sensor Microwave Imager/Sounder (2008–2013) at a 25-km spatial resolution, were projected on the MODIS grid (4.64 km) using a nearest neighbor approach for consistency with the chl *a* concentration grid. Pixels with a chl *a* concentration that exhibited sea ice concentration greater than 10% were discarded to limit the contamination of the ocean color signal by the presence of sea ice. *Ice-adjacency effect* and subpixel ice floes can introduce bias in the remote sensing open-ocean reflectance collected in the vicinity of the ice edge, leading to an underestimation or an overestimation of the chl *a* concentration (Bélanger et al., 2007). However, the use of median instead of arithmetic mean when computing the average chl *a* concentration at a given pixel during the PSB minimizes these effects by removing possible outliers from the analysis.

In addition to chl *a*, PP rates (see below), and sea ice concentrations, daily photosynthetically available radiation (PAR) just beneath the sea surface was computed using MODIS atmospheric products, that is, ozone concentration, cloud optical thickness, and the sun zenith angle, which were used as inputs to lookup tables generated using the Santa Barbara DISORT Atmospheric Radiative Transfer (SBDART) model of Ricchiazzi et al. (1998; see Laliberté et al., 2016, and Ayyala Somayajula et al., 2018, for details on the method and the performance of the PAR model).

2.2. Definition of a PSB Using Ocean Color Data

According to in situ observations, PSBs usually last from 1 to 3 weeks without an extra supply of nutrients resulting, for instance, from an upwelling event (Niebauer et al., 1990; Niebauer & Alexander, 1985). Two criteria used in previous studies (Pabi et al., 2008; Perrette et al., 2011), based on sea ice and chl *a* concentrations thresholds, were used in the current study to operationally set the occurrence of a PSB in a given location in a given year: (1) it takes place during the earliest 20 consecutive days when sea ice concentration is always below 10%, and (2) within this 3-week period, there must be at least one chl *a* concentration observation



Figure 1. Latitudinal distribution in phytoplankton spring blooms (PSBs) of median chlorophyll *a* (chl *a*) concentration (a) and median daily rates of primary productivity (PP) (b) in the open water of the Arctic Ocean between 2003 and 2013.



Figure 2. Estimates of annual trends in daily flux of primary productivity (PP; in mg C m⁻² day⁻¹ year⁻¹) during the phytoplankton spring bloom (PSB) determined from satellite ocean color data from 2003 to 2013 for 40 geographic subsectors of the Arctic Ocean. Solid colored areas correspond to slopes with *p* value ≤ 0.05 , and black stars on solid colored areas correspond to slopes with 0.05 < p value ≤ 0.1 . Striped colored areas denote slopes with a lack of statistical significance (*p* value > 0.1). Black triangles correspond to subsectors located in the central basin (see the supporting information, Table S1). Green pixels correspond to new PSBs observed only since 2010. Black indicates pixels that had sea ice concentration always >10% during the 2003–2013 time period. White areas below 70°N are not included in the analysis as our study is performed between 70° and 90°N. White areas above 70°N correspond to subsectors with not enough observations (i.e., 4 years) to perform the trend analysis.

greater or equal to 0.5 mg/m^3 . The above criteria discarded cases where sea ice concentration re-increased because of growth of new sea ice or the occurrence of drifting of ice by currents and winds. It is also important to keep in mind that using these criteria, blooms that started under sea ice or even blooms in the declining phase are included in our analysis only if the 0.5 mg/m³ chl *a* threshold is met during the open-water 3-week bloom period. Such situations are likely to happen more often at lower latitudes. Our analysis was also conducted for sea ice concentration thresholds of 25% and 50% to determine the beginning of the PSB with no significant changes to the final results.

2.3. Satellite Data Binning and Time Series Analysis

We performed our analyses on all pixels located between 70° and 90°N. First, we calculated the median chl *a* concentrations observed during the PSB and defined according to the above criteria for each pixel. Second, daily rates of PP observed during the PSB were computed using the PP model of Bélanger et al. (2013) as implemented in Benoît-Gagné et al. (2017). Evaluation of the model showed consistent results when compared to in situ measurements (Lee et al., 2015). Because the currency of our model is carbon, the reported PP values may not fully reflect variations in actual growth, which is highly constrained by nutrients. We also calculated the medians of the daily rates of PP and PAR observed during the PSB for each pixel. The day of ice breakup was also recorded for each pixel, according to the above criteria (see section 2.2).

To study the latitudinal progression of the PSB, the medians of all the available median chl *a* concentrations (Figure 1a) and median daily rates of PP (Figure 1b) computed for each pixel were retrieved across all longitudes (between 180°W and 180°E) for 0.5° latitudinal bins from 70 to 90°N. The median was selected over the mean to decrease the influence of possible outliers, as suspicious very high chl *a* concentrations (>200 mg/m³) can be found close to the sea ice edge.

We also computed the annual trends of median daily rates of PP observed during the PSB for eight different sectors of the Arctic Ocean (Figure 2; see the supporting information, Figure S1) as described in Pabi et al. (2008):





Figure 3. Daily rates of photosynthetically available radiation (PAR) observed during the phytoplankton spring bloom (PSB) (a) and timing of ice breakup (b). Daily rates of PAR and timing of ice breakup were retrieved over all longitudes between 2003 and 2013 for the Arctic Ocean and combined over a 2- or 3-year time period (2003–2005, 2006–2008, 2009–2011, and 2012–2013). The dashed red line represents the day of the summer solstice (the 172th day of the year, 21 June).

the Chukchi (180 to 160°W), Beaufort (160 to 100°W), Barents (15 to 55°E), Kara (55 to 105°E), Laptev (105 to 150°E), Siberia (150° to 180°E), Baffin (100° to 45°W), and Greenland (45°W to 15°E) sectors (see the supporting information, Figure S2). These eight sectors were further divided in 3° latitudinal bins (i.e., 70–73°N, 73–76°N, 76–79°N, 79–82°N, and 82–85°N) to produce a total of 40 subsectors. This approach was chosen to separate the time series analysis between cases of new areas of open waters at the highest latitudes and areas where seasonal ice-free water always existed between 2003 and 2013. In addition, it allowed us also to discriminate the subsectors located in the central basin from other subsectors (see the supporting information, Table S1). We proceeded by calculating the median of all the available pixels of median daily rates of PP observed during the PSB for all subsectors and for each year between 2003 and 2013. Only subsectors with at least 4 years of observations of PSBs were used in the analysis, and 28 out of the 40 subsectors met this criterion. We computed the trends as the slope of the linear regression of the annual median versus year. The linear regression was weighed by the number of pixels (i.e., PSBs) available in each subsector for each year. The number of observations for each subsector is presented in Figure S3 (see the supporting information).

We also computed for each pixel the median daily rates of PAR observed during the PSB (Figure 3a) and day of ice breakup (Figure 3b). To minimize some of the interannual variations without aliasing trends, we calculated the median of median daily rates of PAR and timing of ice breakup over a 2- or 3-year time period (2003–2005, 2006–2008, 2009–2011, and 2012–2013).

The annual northernmost latitude, Lat_{max} , reached by the PSB, was derived using only pixels with at least four observations of chl *a* concentrations that matched the criteria defined above. A linear regression of Lat_{max} versus year was performed to detect any trend in the annual maximum latitude of the PSB. The same analysis was performed without the criteria of chl *a* (i.e., chl *a* concentration > 0.5 mg/m³) to obtain information on new ice-free area that did not develop a bloom.

3. Results and Discussion

3.1. Spatial and Temporal Variabilities of Biomass and Daily Rates of PP of PSBs

Chl *a* concentrations and daily rates of PP during the PSB increased at latitudes greater than 75°N from 2003 to 2013 in the Arctic Ocean (Figures 1a and 1b). Reduction in chl *a* concentration observed around 72° to 75°N is likely due to a reduction in nutrients either explained by the occurrence of under-ice blooms or an increase in stratification at these latitudes. We also observed low chl *a* concentrations and daily rates of PP during the

PSB above 81°N in the Arctic basin where daily PAR is low (Figure 3a), and low concentrations of nutrients prevail all year long (Codispoti et al., 2013). Regarding the spatial distribution of PSBs, there was a strong variability in the total number of annual PSBs observed in the Arctic Ocean over the 2003–2013 time period (see the supporting information, Figure S4). A small number of annual PSBs are recorded in northern areas of the Arctic Ocean (>75°N), which were more covered by sea ice early in the time series. Moreover, at pan-Arctic scale, there was a strong variability in the number of observations (i.e., chl *a* concentrations) for a given pixel during the PSB (see the supporting information, Figure S5). The frequent occurrence of clouds or low-lying fog just after ice melt, especially during late summer and early fall, limited the number of available chl *a* data and made the ocean color observations patchy (Perrette et al., 2011).

We observed a strong regional variability of the PP trends in the PSB within the eight different sectors of the Arctic Ocean. Positive and significant annual trends in PP during the PSB were found in 11 (*p* value ≤ 0.05) and 3 (0.05 < p value ≤ 0.1) out of the 28 subsectors with sufficient observations from 2003 to 2013. The highest increase in PP was observed in the Barents (53.54 mg C m⁻² day⁻¹ year⁻¹ between 73° and 76°N) and Kara (75.70 mg C m⁻² day⁻¹ year⁻¹ between 82° and 85°N) subsectors (Figure 2). The Chukchi, Beaufort, Baffin, and Greenland sectors showed the lowest trends of annual PP of about 15 mg C m⁻² day⁻¹ year⁻¹ over the 2003–2013 time period and for different ranges of latitudes between 70° and 82°N (Figure 2). Overall in the Arctic Ocean, we observed a significant increase in PP during the PSB of about 31% between 2003 and 2013 (an annual increasing trend of 21.43 mg C m⁻² day⁻¹ year⁻¹, *p* value <0.01, *R*² = 0.65; see the supporting information, Figure S1). Moreover, the spatial distribution of chl *a* concentration shows that high values occur mostly on the shelves compared to the central basin where the waters are considered as oligotrophic (Ardyna et al., 2011; Coupel et al., 2011, 2015; see the supporting information, Figure S6).

The sea ice breakup was observed later in the season at latitudes above 80°N in the Arctic Ocean (day of the year 195–250 on average, i.e., mid-July to early September) compared to lower latitudes (day of the year 170–210 on average, i.e., mid-June to late July). We found positive and significant annual trends in daily PAR beneath the sea surface during the PSB in four out of the 28 defined subsectors of the Arctic Ocean with sufficient observations from 2003 to 2013, such as the Barents, Kara, and Laptev sectors (see the supporting information, Figure S7). This can be explained by the fact that the breakup happened earlier in the season and closer to the summer solstice (Figure 3b; i.e., at the maximum of daily solar radiation, 21 June), resulting in a longer photoperiod and in an increase of PAR at these latitudes.

Our results suggest that for some subsectors located in the Barents, Kara, and Laptev sectors, an increase in daily PAR due to an earlier ice breakup contributes (i) to an increase in the accumulation of phytoplankton biomass due to favorable bloom dynamics (Behrenfeld et al., 2016) and (ii) to an increase in the daily rates of PP during the open-water PSB observed between 2003 and 2013. This observation is also in accordance with the study of Zhang et al. (2010) who showed concomitant increases in PAR and marine PP in the Arctic Ocean between 1988 and 2007 with a coupled 3-D pan-Arctic model. The ongoing shift in the timing of ice breakup has therefore important consequences for PSB PP in the Arctic. However, other factors than PAR that could affect the phytoplankton bloom, such as nutrient availability and its potential replenishment in the water column, could explain the increase in daily rates of PP during the open-water PSB for other subsectors and should be examined in the future (Popova et al., 2010; Rainville et al., 2011; Tremblay et al., 2015; Tremblay & Gagon, 2009). The observed spatially heterogeneous responses of the annual trends of PP suggest a dependence of PP on multiple confounding factors driving the Arctic ecosystem dynamics that needs to be better understood and characterized.

A mismatch between the spring bloom burst of phytoplankton and the activity of zooplankton grazers resuming winter diapause could also explain this increase in the daily rates of PP during the open-water PSB, since the life cycles of phytoplankton and zooplankton grazers are tightly linked to the timing of the blooms (Behrenfeld et al., 2016; Ji et al., 2013; Leu et al., 2011; Søreide et al., 2010). The PSB is usually followed by a long-delayed peak in zooplankton biomass, dominated by large copepods of the genus *Calanus* that overwinters at depth (Ashjian et al., 2003; Campbell et al., 2009). Because of a low grazer biomass and activity at the onset of the PSB, much of the PP may not be consumed in the water column, as it has been documented, for example, in the Chukchi and Beaufort Seas where more than half of the PP may not be consumed by the zooplankton (Campbell et al., 2009; Sherr et al., 2009).





Figure 4. Highest latitude reached by phytoplankton spring blooms (PSBs) in the open water of the Arctic Ocean from 2003 to 2013. The red dots and red regression line refer to the analysis with the pixels where chl *a* concentration (at least four observations, any concentration) was observed within the 20 consecutive days of ice breakup. The black dots and black regression line show the same, but only when at least 1 value of chl *a* concentration went beyond the threshold of 0.5 mg/m³ and defined the beginning of the PSB. The blue dots and blue regression line refer to the maximum latitude where at least one ice-free pixel (i.e., sea ice concentration below 10% during 20 consecutive days) was observed. Trends computed as the slope of the linear regression of highest latitude versus year are weighed by the number of available observations (i.e., pixels) between 2003 and 2013 (slopes of the linear regression, *p* values, and coefficients of determination R^2).

The increasing PP in PSBs might have contributed to a significant fraction of increased open ocean PP that has notably been observed in recent Arctic studies (Arrigo & van Dijken, 2015; Bélanger et al., 2013; Pabi et al., 2008). It is also important to note that these observations do not account for PP that could occur within or beneath sea ice cover or for PP located at the subsurface chlorophyll maximum. Major phytoplankton under-ice blooms have recently been observed in different regions of the Arctic Ocean (Arrigo et al., 2012; Fortier et al., 2002; Mundy et al., 2009). The occurrence of such under-ice blooms, invisible to satellite remote sensing, may increase in the future in an Arctic Ocean dominated by thinner first year ice (Lowry et al., 2014; Mundy et al., 2009).

3.2. Northward Expansion of PSB Estimates

Lat_{max} reached by the PSB between 2003 and 2013 was recorded for each year. The most northern PSB in the Arctic Ocean was recorded in 2013 at a latitude of 82.23°N compared to the maximum latitude of 80.60°N reached by the PSB in 2003 (Figure 4). The slope of the linear regression of the annual maximum latitude reached by the PSB on time (in year) corresponds to a northward progression of the PSB by about 1° per decade for the Arctic Ocean (*p* value < 0.05, $R^2 = 0.37$). This result is mostly driven by the Barents and Kara sectors where a clear northward progression of these blooms is observed since 2003 at a rate of about 1.1° (*p* value < 0.05, $R^2 = 0.42$) and 1.2° (*p* value < 0.05, $R^2 = 0.54$) per decade, respectively (see the supporting information, Table S2). In addition, most of the recorded PSBs since 2010 occurred above 80°N and were located in the Barents, Kara, and Laptev sectors (Figure 2). Trends in other sectors of the Arctic Ocean were not found to be statistically significant (*p* value > 0.05; see

the supporting information, Table S2). Interestingly, we observed a negative trend of about 2.4° per decade (*p* value = 0.03, R^2 = 0.42), a southward progression of PSBs, in the Greenland sector (see the supporting information, Table S2) where the pixels recorded for the analysis were mostly located in the Fram Strait.

At the highest latitude (i.e., >80°N), the PSB develops around early September (on average the 250th day of year, result not shown). It is noteworthy that the positive trend of the linear regression of the annual maximum latitude reached by the open water is twice as fast as the northward progression of the PSB with a value of 2.1° per decade (*p* value < 0.05, $R^2 = 0.46$, Figure 4, and see the supporting information, Table S2). This may suggest, if nutrients are not the limiting factor, that the ice breakup takes place too late in the year in the most northern regions of the Arctic Ocean, when sun elevation is too low to provide the required energy for phytoplankton to bloom. If ice breakup occurs earlier in the future, the minimum light requirement could be met and PSBs could eventually happen and spread over 82°N.

The northward progression of the PSB observed in the Barents and Kara sectors likely results from an earlier retreat of the ice cover driven by the increased input of warm Atlantic waters through Fram Strait (Årthun et al., 2012; Oziel et al., 2016; Spielhagen et al., 2011). The southward progression of the PSB observed in the Greenland sector could likely be related to the increasing export of sea ice from the Arctic Ocean through Fram Strait preventing northward expansion of PSBs in this area (Halvorsen et al., 2015; Smedsrud et al., 2017). The occurrence of PSBs in the central Arctic basin is also a significant finding given that these remote waters are defined as oligotrophic (Ardyna et al., 2011; Coupel et al., 2011, 2015). Low nutrient concentrations limiting the PP in spring and summer and resulting from the strong stratification of the water column all year long have been documented in this region even in wintertime (Codispoti et al., 2013).

Important and significant changes in annual PSB PP and northward progression of PSBs were observed in the Arctic Ocean, especially in the Barents and Kara sectors. The increasing PP of PSBs and their occurrence in the central basin challenge our understanding of the dynamics of the marine ecosystem in this region and the future role of the Arctic Ocean as a source or a sink of carbon for the global climate. With the decline of the ice cover, competition for light and nutrients could happen between under-ice blooms and PSBs and impact the dynamics of the marine ecosystem. Moreover, in a possibly future Arctic ice-free summer, our

results suggest that PSBs could extend in the central basin, but the PP may likely remain low due to an increased sea ice melting and a subsequent intensified stratification of the water column, preventing nutrient-rich deep waters to be mixed with surface waters (Coupel et al., 2015; Slagstad et al., 2011). On the contrary, storms and associated high winds could trigger strong vertical mixing in an ice-free ocean, increase the nutrient supply in the upper water column, and favor the occurrence of PSBs (Ardyna et al., 2014; Mundy et al., 2009). The impact of PSBs on higher trophic levels (Leu et al., 2011; Post, 2017; Richardson, 2008) and the subsequent export fluxes of organic matter (Wassmann & Reigstad, 2011) remain also to be determined (Yool et al., 2015).

References

- Ardyna, M., Babin, M., Gosselin, M., Devred, E., Bélanger, S., Matsuoka, A., & Tremblay, J. É. (2013). Parameterization of vertical chlorophyll *a* in the Arctic Ocean: Impact of the subsurface chlorophyll maximum on regional, seasonal and annual primary production estimates. *Biogeosciences Discussions*, 10(1), 1345–1399. https://doi.org/10.5194/bgd-10-1345-2013
- Ardyna, M., Babin, M., Gosselin, M., Devred, E., Rainville, L., & Tremblay, J.-É. (2014). Recent Arctic Ocean sea ice loss triggers novel fall phytoplankton blooms. *Geophysical Research Letters*, 41, 6207–6212. https://doi.org/10.1002/2014GL061047
- Ardyna, M., Gosselin, M., Michel, C., Poulin, M., & Tremblay, J. (2011). Environmental forcing of phytoplankton community structure and function in the Canadian High Arctic: Contrasting oligotrophic and eutrophic regions. *Marine Ecology Progress Series*, 442, 37–57. https://doi.org/10.3354/meps09378
- Arrigo, K. R., Matrai, P. A., & van Dijken, G. L. (2011). Primary productivity in the Arctic Ocean: Impacts of complex optical properties and subsurface chlorophyll maxima on large-scale estimates. *Journal of Geophysical Research*, 116, C11022. https://doi.org/10.1029/2011JC007273
- Arrigo, K. R., Perovich, D. K., Pickart, R. S., Brown, Z. W., van Dijken, G. L., Lowry, K. E., et al. (2012). Massive phytoplankton blooms under Arctic sea ice. *Science*, *336*(6087), 1408. https://doi.org/10.1126/science.1215065
- Arrigo, K. R., van Dijken, G., & Pabi, S. (2008). Impact of a shrinking Arctic ice cover on marine primary production. *Geophysical Research Letters*, 35, L19603. https://doi.org/10.1029/2008GL035028
- Arrigo, K. R., & van Dijken, G. L. (2011). Secular trends in Arctic Ocean net primary production. Journal of Geophysical Research, 116, C09011. https://doi.org/10.1029/2011JC007151
- Arrigo, K. R., & van Dijken, G. L. (2015). Continued increases in Arctic Ocean primary production. Progress in Oceanography, 136, 60–70. https:// doi.org/10.1016/j.pocean.2015.05.002
- Årthun, M., Eldevik, T., Smedsrud, L. H., Skagseth, Ø., & Ingvaldsen, R. B. (2012). Quantifying the influence of Atlantic heat on Barents sea ice variability and retreat. *Journal of Climate*, 25(13), 4736–4743. https://doi.org/10.1175/JCLI-D-11-00466.1
- Ashjian, C. J., Campbell, R. G., Welch, H. E., Butler, M., & Van Keuren, D. (2003). Annual cycle in abundance, distribution, and size in relation to hydrography of important copepod species in the western Arctic Ocean. *Deep Sea Research Part I: Oceanographic Research Papers*, 50(10–11), 1235–1261. https://doi.org/10.1016/S0967-0637(03)00129-8
- Assmy, P., Fernandez-Mendez, M., Duarte, P., Meyer, A., Randelhoff, A., Mundy, C. J., et al. (2017). Leads in Arctic pack ice enable early phytoplankton blooms below snow-covered sea ice. *Scientific Reports*, 7(1), 40850. https://doi.org/10.1038/srep40850
- Ayyala Somayajula, S., Devred, E., Bélanger, S., Antoine, D., Vellucci, V., & Babin, M. (2018). Evaluation of sea-surface photosynthetically available radiation algorithms under various sky conditions and solar elevations. *Applied Optics*, 57, 3088–3105. https://doi.org/10.1364/ AO.57.003088
- Babin, M., Bélanger, S., Ellingsen, I., Forest, A., Le Fouest, V., Lacour, T., et al. (2015). Estimation of primary production in the Arctic Ocean using ocean colour remote sensing and coupled physical–biological models: Strengths, limitations and how they compare. *Progress in Oceanography*, 139, 197–220. https://doi.org/10.1016/j.pocean.2015.08.008
- Behrenfeld, M. J., Hu, Y., O'Malley, R. T., Boss, E. S., Hostetler, C. A., Siegel, D. A., et al. (2016). Annual boom-bust cycles of polar phytoplankton biomass revealed by space-based lidar. *Nature Geoscience*, *10*, 118–122. https://doi.org/10.1038/ngeo2861
- Bélanger, S., Babin, M., & Tremblay, J. É. (2013). Increasing cloudiness in Arctic damps the increase in phytoplankton primary production due to sea ice receding. *Biogeosciences*, 10(6), 4087–4101. https://doi.org/10.5194/bg-10-4087-2013
- Bélanger, S., Ehn, J. K., & Babin, M. (2007). Impact of sea ice on the retrieval of water-leaving reflectance, chlorophyll a concentration and inherent optical properties from satellite ocean color data. *Remote Sensing of Environment*, 111(1), 51–68. https://doi.org/10.1016/j. rse.2007.03.013
- Ben Mustapha, S., Bélanger, S., & Larouche, P. (2012). Evaluation of ocean color algorithms in the southeastern Beaufort Sea, Canadian Arctic: New parameterization using SeaWiFS, MODIS, and MERIS spectral bands. *Canadian Journal of Remote Sensing*, 38(5), 1–22.
- Benoît-Gagné, M., Devred, E., Dessailly, D., Bélanger, S., & Babin, M. (2017). PPv0: A primary productivity algorithm implementation from surface chlorophyll-a data above 45 degrees North (release PPv0.36.0.0). Zenodo.
- Campbell, R. G., Sherr, E. B., Ashjian, C. J., Plourde, S., Sherr, B. F., Hill, V., & Stockwell, D. A. (2009). Mesozooplankton prey preference and grazing impact in the western Arctic Ocean. Deep Sea Research Part II: Topical Studies in Oceanography, 56(17), 1274–1289. https://doi.org/ 10.1016/j.dsr2.2008.10.027
- Codispoti, L. A., Kelly, V., Thessen, A., Matrai, P., Suttles, S., Hill, V., et al. (2013). Synthesis of primary production in the Arctic Ocean: Ill. Nitrate and phosphate based estimates of net community production. *Progress in Oceanography*, *110*, 126–150. https://doi.org/10.1016/j. pocean.2012.11.006
- Coupel, P., Jin, H. Y., Ruiz-Pino, D., Chen, J. F., Lee, S. H., Li, H. L., et al. (2011). Phytoplankton distribution in the western Arctic Ocean during a summer of exceptional ice retreat. *Biogeosciences Discussions*, 8(4), 6919–6970. https://doi.org/10.5194/bgd-8-6919-2011
- Coupel, P., Ruiz-Pino, D., Sicre, M. A., Chen, J. F., Lee, S. H., Schiffrine, N., et al. (2015). The impact of freshening on phytoplankton production in the Pacific Arctic Ocean. Progress in Oceanography, 131, 113–125. https://doi.org/10.1016/j.pocean.2014.12.003
- Fortier, M., Fortier, L., Michel, C., & Legendre, L. (2002). Climatic and biological forcing of the vertical flux of biogenic particles under seasonal Arctic sea ice. *Marine Ecology Progress Series*, 225, 1–16. https://doi.org/10.3354/meps225001
- Haas, C., Pfaffling, A., Hendricks, S., Rabenstein, L., Etienne, J.-L., & Rigor, I. (2008). Reduced ice thickness in Arctic transpolar drift favors rapid ice retreat. *Geophysical Research Letters*, 35, L17501. https://doi.org/10.1029/2008GL034457
- Halvorsen, M. H., Smedsrud, L. H., Zhang, R., & Kloster, K. (2015). Fram Strait spring ice export and September Arctic sea ice. *The Cryosphere Discussions*, 9(4), 4205–4235. https://doi.org/10.5194/tcd-9-4205-2015

Acknowledgments

MODIS data were made available by the NASA's Ocean Color Processing Group at http://oceandata.sci.gsfc.nasa.gov. Sea ice concentration data were provided by the National Snow and Ice Data Center from their website http://nsidc.org. We gratefully thank M. Benoît-Gagné, P. Massicotte, L. Oziel, and E. Rehm for programming assistance and helpful comments on the manuscript, and D. Christiansen-Stowe for language support. We also thank two anonymous reviewers for constructive and insightful reviews. S. R. received a postgraduate scholarship from the Canada Excellence Research Chair (CERC) in Remote Sensing of Canada's New Arctic Frontier and stipends from Québec-Océan and ArcticNet. This is a contribution to the research programs of the CERC in Remote Sensing of Canada's New Arctic Frontier, Takuvik Joint International Laboratory, ArcticNet, and Ouébec-Océan.

- Ji, R., Jin, M., & Varpe, O. (2013). Sea ice phenology and timing of primary production pulses in the Arctic Ocean. *Global Change Biology*, 19(3), 734–741. https://doi.org/10.1111/gcb.12074
- Kahru, M., Brotas, V., Manzano-Sarabia, M., & Mitchell, B. G. (2011). Are phytoplankton blooms occurring earlier in the Arctic? Global Change Biology, 17(4), 1733–1739. https://doi.org/10.1111/j.1365-2486.2010.02312.x
- Kahru, M., Lee, Z., Mitchell, B. G., & Nevison, C. D. (2016). Effects of sea ice cover on satellite-detected primary production in the Arctic Ocean. Biology Letters, 12(11). https://doi.org/10.1098/rsbl.2016.0223
- Kröncke, I., Vanreusel, A., Vincx, M., Wollenburg, J., Mackensen, A., Liebezeit, G., & Behrends, B. (2000). Different benthic size-compartments and their relationship to sediment chemistry in the deep Eurasian Arctic Ocean. *Marine Ecology Progress Series*, 199, 31–41. https://doi.org/ 10.3354/meps199031
- Laliberté, J., Bélanger, S., & Frouin, R. (2016). Evaluation of satellite-based algorithms to estimate photosynthetically available radiation (PAR) reaching the ocean surface at high northern latitudes. *Remote Sensing of Environment*, 184, 199–211. https://doi.org/10.1016/j. rse.2016.06.014
- Lee, Y. J., Matrai, P. A., Friedrichs, M. A. M., Saba, V. S., Antoine, D., Ardyna, M., et al. (2015). An assessment of phytoplankton primary productivity in the Arctic Ocean from satellite ocean color/in situ chlorophyll-abased models. *Journal of Geophysical Research: Oceans, 120*, 6508–6541. https://doi.org/10.1002/2015JC011018
- Leu, E., Søreide, J. E., Hessen, D. O., Falk-Petersen, S., & Berge, J. (2011). Consequences of changing sea-ice cover for primary and secondary producers in the European Arctic shelf seas: Timing, quantity, and quality. *Progress in Oceanography*, 90(1–4), 18–32. https://doi.org/ 10.1016/j.pocean.2011.02.004
- Lowry, K. E., van Dijken, G. L., & Arrigo, K. R. (2014). Evidence of under-ice phytoplankton blooms in the Chukchi Sea from 1998 to 2012. Deep Sea Research Part II: Topical Studies in Oceanography, 105, 105–117. https://doi.org/10.1016/j.dsr2.2014.03.013
- Maritorena, S., & Siegel, D. A. (2005). Consistent merging of satellite ocean color data sets using a bio-optical model. *Remote Sensing of Environment*, 94(4), 429–440. https://doi.org/10.1016/j.rse.2004.08.014
- Martin, J., Tremblay, J., Gagnon, J., Tremblay, G., Lapoussière, A., Jose, C., et al. (2010). Prevalence, structure and properties of subsurface chlorophyll maxima in Canadian Arctic waters. *Marine Ecology Progress Series*, 412, 69–84. https://doi.org/10.3354/meps08666
- Matrai, P. A., Olson, E., Suttles, S., Hill, V., Codispoti, L. A., Light, B., & Steele, M. (2013). Synthesis of primary production in the Arctic Ocean: I. Surface waters, 1954–2007. Progress in Oceanography, 110, 93–106. https://doi.org/10.1016/j.pocean.2012.11.004
- Mundy, C. J., Gosselin, M., Ehn, J., Gratton, Y., Rossnagel, A., Barber, D. G., et al. (2009). Contribution of under-ice primary production to an iceedge upwelling phytoplankton bloom in the Canadian Beaufort Sea. *Geophysical Research Letters*, 36, L17601. https://doi.org/10.1029/ 2009GL038837
- Mundy, C. J., Gosselin, M., Gratton, Y., Brown, K., Galindo, V., Campbell, K., et al. (2014). Role of environmental factors on phytoplankton bloom initiation under landfast sea ice in Resolute Passage, Canada. *Marine Ecology Progress Series*, 497, 39–49. https://doi.org/10.3354/ meps10587
- Niebauer, H. J., & Alexander, V. (1985). Oceanographic frontal structure and biological production at an ice edge. *Continental Shelf Research*, 4(4), 367–388. https://doi.org/10.1016/0278-4343(85)90001-9

Niebauer, H. J., Alexander, V., & Henrichs, S. (1990). Physical and biological oceanographic interaction in the spring bloom at the Bering Sea marginal ice edge zone. Journal of Geophysical Research, 95(C12), 22229. https://doi.org/10.1029/JC095iC12p22229

- Oziel, L., Sirven, J., & Gascard, J. C. (2016). The Barents Sea frontal zones and water masses variability (1980–2011). Ocean Science, 12(1), 169–184. https://doi.org/10.5194/os-12-169-2016
- Pabi, S., van Dijken, G. L., & Arrigo, K. R. (2008). Primary production in the Arctic Ocean, 1998–2006. Journal of Geophysical Research, 113, C08005. https://doi.org/10.1029/2007JC004578
- Perrette, M., Yool, A., Quartly, G. D., & Popova, E. E. (2011). Near-ubiquity of ice-edge blooms in the Arctic. *Biogeosciences*, 8(2), 515–524. https://doi.org/10.5194/bg-8-515-2011
- Petrenko, D., Pozdnyakov, D., Johannessen, J., Counillon, F., & Sychov, V. (2013). Satellite-derived multi-year trend in primary production in the Arctic Ocean. International Journal of Remote Sensing, 34(11), 3903–3937. https://doi.org/10.1080/01431161.2012.762698
- Popova, E. E., Yool, A., Coward, A. C., Aksenov, Y. K., Alderson, S. G., de Cuevas, B. A., & Anderson, T. R. (2010). Control of primary production in the Arctic by nutrients and light: Insights from a high resolution ocean general circulation model. *Biogeosciences Discussions*, 7(4), 5557–5620. https://doi.org/10.5194/bgd-7-5557-2010
- Post, E. (2017). Implications of earlier sea ice melt for phenological cascades in arctic marine food webs. *Food Webs*, *13*, 60–66. https://doi. org/10.1016/j.fooweb.2016.11.002
- Post, E., Bhatt, U. S., Bitz, C. M., Brodie, J. F., Fulton, T. L., Hebblewhite, M., et al. (2013). Ecological consequences of sea-ice decline. Science, 341(6145), 519–524. https://doi.org/10.1126/science.1235225
- Rainville, L., Lee, C., & Woodgate, R. (2011). Impact of wind-driven mixing in the Arctic Ocean. Oceanography, 24(3), 136–145. https://doi.org/ 10.5670/oceanog.2011.65
- Ricchiazzi, P., Yang, S., Gautier, C., & Sowie, D. (1998). SBDART: A research and teaching software tool for plane-parallel radiative transfer in the Earth's atmosphere. *Bulletin of the American Meteorological Society*, 79(10), 2101–2114. https://doi.org/10.1175/1520-0477(1998)079<2101:SARATS>2.0 CO:2
- Richardson, A. J. (2008). In hot water: Zooplankton and climate change. ICES Journal of Marine Science, 65(3), 279–295. https://doi.org/ 10.1093/icesjms/fsn028
- Sakshaug, E. (2004). Primary and secondary production in the Arctic seas. In R. Stein & R. W. MacDonald (Eds.), The Organic Carbon Cycle in the Arctic Ocean (pp. 57–81). Berlin: Springer-Verlag.
- Sakshaug, E., & Skjoldal, H. R. (1989). Life at the ice edge. Ambio, 18(1), 60-67.
- Serreze, M. C., Holland, M. M., & Stroeve, J. (2007). Perspectives on the Arctic's shrinking sea-ice cover. *Science*, *315*(5818), 1533–1536. https://doi.org/10.1126/science.1139426
- Sherr, E., Sherr, B., & Hartz, A. (2009). Microzooplankton grazing impact in the western Arctic Ocean. Deep Sea Research Part II: Topical Studies in Oceanography, 56(17), 1264–1273. https://doi.org/10.1016/j.dsr2.2008.10.036
- Slagstad, D., Ellingsen, I. H., & Wassmann, P. (2011). Evaluating primary and secondary production in an Arctic Ocean void of summer sea ice: An experimental simulation approach. *Progress in Oceanography*, 90(1–4), 117–131. https://doi.org/10.1016/j. pocean.2011.02.009
- Smedsrud, L. H., Halvorsen, M. H., Stroeve, J. C., Zhang, R., & Kloster, K. (2017). Fram Strait sea ice export variability and September Arctic sea ice extent over the last 80 years. *The Cryosphere*, 11(1), 65–79. https://doi.org/10.5194/tc-11-65-2017
- Søreide, J. E., Leu, E. V. A., Berge, J., Graeve, M., & Falk-Petersen, S. (2010). Timing of blooms, algal food quality and *Calanus glacialis* reproduction and growth in a changing Arctic. *Global Change Biology*. https://doi.org/10.1111/j.1365-2486.2010.02175.x

- Spielhagen, R. F., Werner, K., Sorensen, S. A., Zamelczyk, K., Kandiano, E., Budeus, G., et al. (2011). Enhanced modern heat transfer to the Arctic by warm Atlantic water. *Science*, 331(6016), 450–453. https://doi.org/10.1126/science.1197397
- Stroeve, J. C., Serreze, M. C., Holland, M. M., Kay, J. E., Malanik, J., & Barrett, A. P. (2012). The Arctic's rapidly shrinking sea ice cover: A research synthesis. *Climatic Change*, 110(3–4), 1005–1027. https://doi.org/10.1007/s10584-011-0101-1
- Tremblay, J.-É., Anderson, L. G., Matrai, P., Coupel, P., Bélanger, S., Michel, C., & Reigstad, M. (2015). Global and regional drivers of nutrient supply, primary production and CO₂ drawdown in the changing Arctic Ocean. *Progress in Oceanography*, 139, 171–196. https://doi.org/ 10.1016/j.pocean.2015.08.009
- Tremblay, J.-E., & Gagon, J. (2009). The effects of irradiance and nutrient supply on the productivity of Arctic waters: A perspective on climate change.
- Wassmann, P., Carroll, J., & Bellerby, R. G. J. (2008). Carbon flux and ecosystem feedback in the northern Barents Sea in an era of climate change: An introduction. *Deep Sea Research Part II: Topical Studies in Oceanography*, 55(20–21), 2143–2153. https://doi.org/10.1016/j. dsr2.2008.05.025
- Wassmann, P., & Reigstad, M. (2011). Future Arctic Ocean seasonal ice zones and implications for pelagic-benthic coupling. *Oceanography*, 24(3), 220–231. https://doi.org/10.5670/oceanog.2011.74
- Yool, A., Popova, E. E., & Coward, A. C. (2015). Future change in ocean productivity: Is the Arctic the new Atlantic? *Journal of Geophysical Research: Atmospheres, 120, 7771–7790.* https://doi.org/10.1002/2015JC011167
- Zhang, J., Spitz, Y. H., Steele, M., Ashjian, C., Campbell, R., Berline, L., & Matrai, P. (2010). Modeling the impact of declining sea ice on the Arctic marine planktonic ecosystem. *Journal of Geophysical Research*, *115*, C10015. https://doi.org/10.1029/2009JC005387